



Is De-escalated Bisphosphonates Therapy a Suitable Alternative to Standard Dosing in Malignant Tumor Patients With Bone Metastases: A Systematic Review and Meta-Analysis

Qiuhua Luo^{1†}, Peng Men^{2†}, Zhiyong Liu³, Suodi Zhai^{2*} and Mingyan Jiang^{1*}

OPEN ACCESS

Edited by:

José Bines, Brazilian National Cancer Institute (INCA), Brazil

Reviewed by:

Ching-Hung Lin, National Taiwan University, Taiwan Tomás Pascual Martinez, Hospital Clínic of Barcelona, Spain

*Correspondence:

Suodi Zhai zhaisuodi@163.com Mingyan Jiang ydyyyxb@163.com

[†]These authors have contributed equally to this work

Specialty section:

This article was submitted to Women's Cancer, a section of the journal Frontiers in Oncology

Received: 04 March 2019 Accepted: 31 July 2019 Published: 14 August 2019

Citation:

Luo Q, Men P, Liu Z, Zhai S and Jiang M (2019) Is De-escalated Bisphosphonates Therapy a Suitable Alternative to Standard Dosing in Malignant Tumor Patients With Bone Metastases: A Systematic Review and Meta-Analysis. Front. Oncol. 9:774. doi: 10.3389/fonc.2019.00774 ¹ Department of Pharmacy, The First Affiliated Hospital of China Medical University, Shenyang, China, ² Department of Pharmacy, Peking University Third Hospital, Beijing, China, ³ Liaoning Center for Drug and Device Evaluation and Monitoring, Shenyang, China

Background: Previous studies have preliminarily identified the non-inferior efficacy for reducing skeletal-related event (SRE) rates between de-escalated (Q12w) and standard (Q3-4w) bone-targeting agents therapy in malignant tumor patients with bone metastases. In this study, we aim to make further efforts to analyze whether the de-escalated bisphosphonates (BPs) strategy is a suitable option by comprehensively retrieving and synthesizing state-of-the-art evidence.

Methods: An extensive electronic search for randomized controlled trials (RCTs) comparing a BPs standard strategy with the de-escalated one in patients with bone metastases was performed up to June 2018. Outcomes of interest were general and found individual types of SRE, skeletal morbidity rate (SMR), bone pain, bone turnover biomarkers and adverse events (AEs). Continuous and dichotomous outcomes were summarized by the weighted mean difference (WMD) and risk ratio (RR), respectively, with 95% confidence intervals (CIs).

Results: A total of eight studies, representing six unique trials (involving 3114 patients), were included. Pooled results indicated comparable efficacy on general SRE (RR 0.99, 95% Cl 0.87–1.12; P = 0.86; $l^2 = 0\%$) and SMR (WMD 0.00, 95% Cl -0.02 - 0.03; P = 0.81; $l^2 = 0\%$). However, the rate of surgery involving bones was significantly higher in de-escalated group than standard group (RR 1.92, 95% Cl 1.17–3.15; P = 0.01; $l^2 = 0\%$) among individual types of SRE. Several trials also demonstrated increased levels of C-terminal or N-terminal telopeptide in de-escalated group. Meta-analyses for gastrointestinal disorders, dizziness and back pain showed significant reductions by 27% (RR 0.73, 95% Cl 0.57–0.94; P = 0.01; $l^2 = 0\%$), 48% (RR 0.52 95% Cl 0.32–0.86; P = 0.01; $l^2 = 0\%$), and 29% (RR 0.71, 0.51–0.99; P = 0.04; $l^2 = 0\%$), respectively, compared to the standard therapy.

1

Conclusion: For malignant tumor patients with bone metastases, a de-escalated BPs strategy is proved to have a better safety profile compared to standard dosing. Although the efficacy is generally comparable on SRE and SMR between the two dosing regimens, trials with long duration and large sample sizes are still warranted to make a solid judgment.

Keywords: bisphosphonates, bone metastases, de-escalated therapy, ONJ (osteonecrosis of the jaws), adverse effect

INTRODUCTION

Bone metastases disease develops as a common clinical problem in malignant tumors. The pathophysiology of bone metastases causes a series of skeletal-related events (SREs), including severe bone pain, pathological fractures, hypercalcemia, and spinal cord compression (1-3). The associated complications caused by bone lesions, in consequence, increase the need for palliative radiotherapy or surgery to bone, which also leads to ongoing symptoms and deterioration of the quality of life (4).

The administration of bisphosphonates (BPs) has been confirmed to potently inhibit the osteoclast-mediated bone resorption, and therefore, delays the onset of skeletal complications (5–8). A recent systematic review of BPs for breast cancer treatment published in the Cochrane Library showed evidence that BPs reduce the risk of developing SREs, delay the median time to an SRE, and appear to reduce bone pain compared to placebo or none BPs therapy, for women with metastatic breast cancer and bone metastases (9).

Currently, monthly BPs therapy is recommended as standard care for patients diagnosed with bone metastases for at least 2 years, based on data derived from the studies of hypercalcaemia of malignancy (10-13). However, several studies have addressed the dosing interval and demonstrated that longer-interval dosing of BPs might maintains a comparable efficacy on reducing the risk of SREs along with reduced side effects. Particularly, it is noteworthy that the cumulative exposure of BPs is associated with significant toxicities, including osteonecrosis of the jaw (ONJ), renal dysfunctions, and gastrointestinal disorders, due to its relatively long half-life and preferential binding and accumulation in bone (14, 15). Their optimal dosing interval has accordingly come into question. On the other hand, as new anticancer agents continue to prolong the life expectancy of patients with malignant tumors, a suitable treatment schedule of BPs is required for a long-term medication to preferably increase the medication compliance (16).

A previous systematic review has explored de-escalated dosing of bone-targeting agents in patients with metastatic breast cancer (17). This systematic review performs a meta-analysis study of SRE, bone turnover biomarkers, and preliminarily identified the non-inferior efficacy between de-escalated (Q12w) and standard (Q3-4w) therapy. However, BP-specific results were not examined separately and individual types of SRE were not examined either. Some drug-related adverse reactions, including nausea, vomiting, abdominal pain, diarrhea, and constipation, that are commonly known to occur during treatment of bone-metastatic malignancies with BPs (18, 19), remain to be further evaluated through meta-analysis. It is therefore of interest to gain a more comprehensive understanding of the efficacy and safety profile of BPs administration in the comparison of different dosing regimens, among a broader population of cancer patients with bone metastases. Thus, we performed a state-ofthe-art systematic review and meta-analysis, to further analyze whether the de-escalation strategy is a suitable option in patients with bone metastases from malignant tumors.

METHODS

The systematic review and meta-analysis were undertaken using a predetermined protocol and reported in accordance with the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) statement (20).

Literature Search Strategy

All relevant studies, describing randomized controlled trials (RCTs) of de-escalated (Q12w) vs. standard (Q3-4w) administration protocols for BPs treatment cycles were sought. An electronic literature search was performed from inception to July 2018 by searching the following databases: PubMed, Embase, and the Cochrane Library. The search strategy for PubMed was provided in **Figure S1**. Additional meeting abstracts were identified from meeting abstracts of American Society of Clinical Oncology, European Society for Medical Oncology, American Association for Cancer Research n and European Association for Cancer Research n end European and unpublished RCTs. We also manually searched the reference lists of relevant studies.

Study Selection and Data Extraction

The eligibility criteria for the systematic review was in accordance with the PICOS (participants, interventions, comparators, outcomes, and study design) approach. RCTs meeting the following criteria were considered for inclusion:

- Participants: patients with malignant tumors who had at least one site of bone involvement.
- Interventions: BPs administrated iv Q12w as a de-escalated arm.
- Comparators: BPs administrated iv Q12w vs. Q3-4w.
- Outcomes: SRE-related outcomes (general SRE, individual types of SRE and time-to-first on-study SRE), skeletal

morbidity rate (SMR), bone pain, changes of bone turnover biomarkers and adverse event (AE).

The screening of titles, abstracts, and full-text references was performed by two reviewers, independently, to identify a set of potentially relevant citations. Data extraction was collected and arranged by researchers using a collection form. Publication sources, intervention details, patient inclusion criteria, and demographics, as well as outcome measures mentioned earlier, were extracted. When detailed full-texts were not available, the relevant contents from abstracts were used for data extraction (but not assessed for risk of bias). Corresponding authors were contacted for data not available within studies, or when outcomes were presented in an unsuitable format for data synthesis.

Risk of Bias

The quality of individual studies was assessed by using the Cochrane Collaboration's risk of bias tool according to five domains as follows: random sequence generation (selection bias); allocation concealment (selection bias); blinding (performance bias and detection bias); incomplete outcome data (attrition bias); and selective reporting (reporting bias). Any discrepancy in the quality of RCTs was resolved by discussions among two reviewers, or by the assistance of a third researcher if necessary.

Data Analysis

Meta-analysis was performed by Review Manager 5.2 software (RevMan, Cochrane, London, UK). The continuous outcome was summarized by WMD, while the dichotomous outcome was summarized by RR with corresponding 95% confidence intervals (CIs). Statistical homogeneity among RCTs was calculated by

using the chi-square test and I^2 -value. The outcome data were synthesized by random-effect models ($I^2 > 50\%$) or fixed-effect models ($I^2 \le 50\%$), respectively, depending on the amount of heterogeneity observed. Forest plots and study-level effect estimates were designed to present various research indicators. If the extracted data was not sufficient for a quantitative metaanalysis, a narrative approach was conducted to summarize the study-specific results.

RESULTS

Selection of Studies

A total of 4,738 citations were retrieved through electronic search from Pubmed, Embase databases and the Cochrane library, from which 1,235 duplicated records were excluded and 3,503 potentially eligible reports were identified by reviewing study titles and abstracts. Based on the inclusion criteria established for the present study, an additional record was obtained from ClinicalTrial.gov. After the full-text screening, a total of eight RCTs consisting of 3114 patients were included, among which one study was closed early as a consequence of slow patient accrual. **Figure 1** provides an overview of the process of study selection.

Study Characteristics

Characteristics of included studies were described in **Table 1**, including study design, interventions, sample size, patient inclusion criteria, outcomes measure, and study duration. One study was published in 2012 (24), three were in 2013 (14, 22, 26), one was in 2014 (21), and the other two were in 2017 (23, 25).



TABLE 1 | Overview of characteristics of included studies.

Study	Study design	Interventions (<i>n</i>)		Sample size (n)	Mean age (years)	Patient inclusion criteria	Outcomes measure	Duration
		De-escalated	Standard					
Addison et al. (21)	Pilot, randomized, non-inferiority trial (data from the pilot, feasibility randomized trial)	01	90 mg pamidronate i.v. Q3-4w (13); concomitant VD ₃ (800-1000 IUs/day) and calcium (1,200–1,500 mg/day)	30	NR	MBC; baseilne serum CTx < 600 ng/L; ≥ 3 months of prior anti-bone resorption therapy	CTx, BAP, BPI, FACT-BP	1 year
Amadori et al. (14) (ZOOM)	Multicenter, prospective, randomized, open-label, non-inferiority trial	4 mg zoledronate i.v. Q12w (209); supplementary calcium (500 mg/day) and VD (400–500 IUs/day)	4 mg zoledronate i.v. Q4w (216); supplementary calcium (500 mg/day) and VD (400–500 IUs/day)	425	Q4w: 59.8 (Median) Q12w: 60.4 (Median)	Stage IV MBC; 12-15 months of prior zoledronate use	SMR, the incidence of SRE/year, time-to-first SRE, bone pain, NTx	1 year
Amir et al. (22) (REFORM)	Randomized feasibility study	90 mg pamidronate i.v. Q12w (19)	90 mg pamidronate i.v. Q3-4w (19)	38	Q3-4w: 55 (Median) Q12w: 60 (Median)	MBC; baseline serum CTx < 600 ng/L; ≥ 3 months of prior BPs use	CTx, BAP, FACT-BP, BPI, SRE	1 year
Himelstein et al. (23)	Randomized, open-label clinical trial	Each zoledronate dose was adjusted for calculated creatinine clearance using actual body weight, i.v., Q12w (911); 500 mg calcium and 400–800 IUs VD/day	Each zoledronate dose was adjusted for calculated creatinine clearance using actual body weight, i.v., Q4w (911); 500 mg calcium and 400–800 IUs VD/day	1822	Q3-4w: 65 (Median) Q12w: 66 (Median)	MBC, MPC, MM; ECOG score of 0 to 12, Cr ≥ 20 mL/min, serum calcium level between 8.0 mg/dL to 11.6 mg/dL; No prior BPs	and kidney dysfunction,	2 year
Coleman et al. (24) (BISMARK)	Open-label, randomized trial	M-ZOL (Q15-16w; Q8-9w or Q3-4w)	S-ZOL (Q3-4w)	289	NR	MBC; No prior BPs	SRE, SMR	2 year
Hortobagyi et al. (25) (OPTIMIZE-2)	Prospective, randomized, double-blind, multicenter clinical trial	4 mg zoledronate i.v. Q12w (203)	4 mg zoledronate i.v. Q4w (200)	416	Q3-4w: 59.2 (Mean) Q12w: 58.6 (Mean) placebo: 60.8 (Mean)	MBC; 10–15 months of prior zoledronate and/or pamidronate use	SRE rate, time-to-first SRE, SMR	1 year
Kuchuk et al. (26)	Updated data from REFORM study	90 mg pamidronate i.v. Q12w (19)	90 mg pamidronate i.v. Q3-4w (19)	38	NR	MBC; baseilne serum CTx < 600 ng/L; ≥3 months of prior BPs use	Correlation between pain scores and CTx, FACT-BP scores	1 year
ClinicalTrials.gov Identifier: NCT00424983 (27)	Randomized, open-label, multi-center, comparative 2-arm trial	4 mg zoledronate i.v. Q12w (9)	4 mg zoledronate i.v. Q4w (9)	18	Q3-4w: 62.0 (Median) Q12w: 60.0 (Median)	MBC, MM; 9 to 12 infusion of zoledronate during the previous year	SRE, SMR, AUC _{0–24h} , CLor	1 year

MBC, metastatic breast cancer; MPC, metastatic prostate cancer; MM, multiple myeloma; SRE, skeletal-related events; SMR, skeletal morbidity rate; CTx, serum C-terminal telopeptide; NTx, urinary N-terminal telopeptide; BAP, alkaline phosphatase; CLcr, creatine clearance; FACT-BP, Functional Assessment of Cancer Therapy Bone Pain; BPI, Brief Pain Inventory; ECOG, Eastern Cooperative Oncology Group; NR, not reported.

De-escalated Bisphosphonates for Bone Metastases

Two studies were of different design in dose administration compared to the other de-escalation studies, as each zoledronate dose was adjusted for calculated creatinine clearance using actual body weight (23) or the dosing interval was based on bone turnover biomarker levels of the individual patient (24). One study was from Phase I clinical trial results (ClinicalTrials.gov Identifier: NCT00424983) (27), which had never been published in a peer-reviewed journal. Because of the limited information given in the trial registry, the detailed data were retrieved from the trial's sponsor (Novartis). One study published updated data from the included studies (27). The BISMARK study closed before reaching the primary endpoint, and data was obtained from updated study reports from the American Society of Clinical Oncology (ASCO) meetings (24). The BPs evaluated within studies were either pamidronate or zoledronate, and the study duration varied between 1 and 2 years.

Patient Characteristics

Patients diagnosed with metastatic breast cancer (MBC), metastatic prostate cancer (MPC) or multiple myeloma (MM) were randomly assigned to a de-escalated dosing or standard dosing of BPs. Three used additional study entry criteria based on low baseline serum C-terminal telopeptide (CTx, < 600 ng/L) (21, 23, 26). One study included Eastern Cooperative Oncology Group (ECOG) performance status score of 0 to 2, calculated creatinine clearance of \geq 30 mL/min and serum calcium level between 8.0 mg/dL and 11.6 mg/dL as inclusion criteria (23). The prior treatment of BPs used before enrollment varied from 0 to 15 months.

Risk of Bias Assessment

The assessment of the risk of bias for individual studies was summarized in **Figure 2**. Five full-text articles were available for risk of bias assessment (14, 22, 23, 25). Two studies were excluded from bias assessment as the study used the same randomized trial data as one of the included studies (21, 26). Four studies were explicit about each allocation concealment and blinding of the outcome assessment (14, 22, 23, 25).

SRE-Related Outcomes General SRE

Studies reported a number of SRE-related outcomes including the SRE rate, SMR, and time-to-first on-study SRE. Five included studies (2,703 patients) investigated the effect of different dosing interval of BPs on the SRE rate (14, 22, 23, 25, 27). A pooled RR of 0.99 (95% CI 0.87–1.12; P = 0.86) indicated that there was no evidence of a significant difference in SRE rate between deescalated and standard arms (**Figure 3**). No heterogeneity was observed amongst the studies ($I^2 = 0\%$). The BISMARK study closed before reaching its primary endpoint, however, it showed no statistically significant difference in the occurrence of SREs between the different intervention groups (24). In addition, two studies reported data of the time of the onset of first SRE. The OPTIMIZE-2 study showed time-to-first SRE between treatment groups was not statistically different (hazard ratio, 1.06; 95% CI, 0.70–1.60; P = 0.79) (25). While the ZOOM study presented that



the median time to first on-study SRE could not be calculated because of the very low event rate (14).

Individual Type of SRE

Studies reported a number of individual types of SRE, including radiation to bone, clinical fractures, spinal cord compression, and surgery involving bone. The most frequently recorded individual type of SRE in both treatment groups was radiation to bone in de-escalated vs. standard group (15.62 vs. 17.71%), followed by pathological fractures (7.97 vs. 6.43%), surgery involving bone (3.90 vs. 2.02%) and spinal cord compression (2.83 vs. 2.11%) (Table S1). A summary RR of 0.88 (95% CI 0.74–1.04; *P* = 0.92; $I^2 = 0\%$) (14, 22, 23, 25, 27), 1.25 (95% CI 0.93–1.69; P =0.14; $I^2 = 0\%$) and 1.34 (95% CI 0.79–2.25; P = 1.34; $I^2 = 0\%$) indicated there was no statistically significant differences between the arms, with respect to radiation to bone, clinical fractures and spinal cord compression, respectively (14, 23, 25). However, comparison in the aspect of surgery involving bone produced a pooled RR of 1.92 (95% CI 1.17-3.15; P = 0.01; $I^2 = 0\%$; **Figure 4**) (14, 23, 25).

Skeletal Morbidity Rates (SMR)

Three studies reported SMR data involved with zoledronate and manifested de-escalated therapy with BPs did not appear to affect an overall SMR (WMD 0.00, 95% CI -0.02 - 0.03; P = 0.81; $I^2 = 0\%$; **Figure 5**) (14, 23, 25). Besides, the BISMARK study, designed by a fundamentally different form as mentioned before, reported SMR of 0.52 (90% CI 0.47–0.58) and 0.72 (90% CI 0.67–0.77) for the standard and bone-biomarker-directed therapy, respectively (24).

Bone Pain

Four studies reported data of bone pain, while differed in the employed assessment tool for pain response. Thus, a metaanalysis was not carried out. The ZOOM study, relying on the Verbal Rating Score (VRS), detected no relevant differences between groups for bone pain at the end of study or analgesic use during follow-up. The median pain at rest and on movement scores were <4 at all points in both groups. The REFORM study and Addison et al. demonstrated that the pain scores remained generally stable over time in both the standard and the deescalated groups, according to validated Brief Pain Inventory (BPI) and Functional Assessment of Cancer Therapy-Bone Pain (FACT-BP). There were no statistically significant differences between groups in cumulative pain scores as measured by BPI (P = 0.21) or by FACT-BP (P = 0.59) (21, 22). Himelstein et al. also used ECOG performance status to evaluate bone pain data, showing the score of 0.025 for every 4 weeks and 0.024 for every 12 weeks. Similarly, the result showed no statistically significant differences at each time point (P = 0.64), as well as mean worst pain within the past 24 h (0.022 vs. 0.021, P = 0.96), mean current pain (0.016 vs. 0.018, P = 0.82), composite pain (0.021 vs. 0.022, P = 0.88), mean relief from pain with treatments or medications (0.009 vs, 0.016, P = 0.59), and mean interference score (0.023 vs, 0.016)0.019, P = 0.68) comparing the de-escalated and standard group (P > 0.001) (23).

AEs

AEs occurred more commonly in the standard group than in the de-escalated group, which might be driven by higher incidence of renal dysfunction, ONJ and gastrointestinal disorders (including nausea, vomiting, abdominal pain, diarrhea, and/or constipation) with more frequent medication. A series of specific toxicities were described in detail by treatment comparison in the form of forest plots (Figure 6 and Figures S2-S5). Data were available from five studies reporting on-study AEs for meta-analysis (Figure 6A). Three studies reported the number of patients with at least one AE, and the result of the meta-analysis showed no evidence of a difference between the arms (RR 0.95, 95% CI 0.87-1.04; P = 0.25) with the random-effect model ($I^2 = 56\%$) (14, 25, 27). For on-study renal dysfunction, comparison of de-escalated and standard treatment groups using meta-analysis produced a pooled RR of 0.69 (95% CI 0.41–1.15; P = 0.16; $I^2 = 0\%$; Figure 6C) (14, 22, 23, 26, 27). For on-study ONJ, comparison of treatment groups produced a pooled RR of 0.56 (95% CI 0.29–1.08; P = 0.10; Figure 6D). No heterogeneity was detected $(I^2 = 0\%)$ (14, 22, 23, 26, 27). Neither renal dysfunction nor ONJ was observed in the REFORM study, probably due to the small number of recruitment. Notably, the de-escalated therapy of BPs reduced the incidence of gastrointestinal disorders by 27% compared to the standard therapy (RR 0.73, 95% CI 0.57–0.94; P = 0.01; $I^2 = 0\%$; **Figure 6B**) (14, 27). The gastrointestinal reactions were further refined into nausea, vomiting, decreased appetite, diarrhea, constipation, and abnominal pain. The numbers of patients who experienced these symptoms were processed for meta-analyses. The results showed a summary RR of 0.84 (95% CI 0.65–1.09; P = 0.19; $I^2 = 0\%$) (14, 25, 27), 0.85 (95% CI 0.60–1.20; P = 0.35; $I^2 = 9\%$) (14, 25, 27), 0.92 (95% CI 0.58–1.46; P = 0.72; $I^2 = 26\%$) (14, 27), 0.83 (95% CI 0.59– 1.17; P = 0.29; $I^2 = 0\%$) (14, 25, 27) and 0.96 (95% CI 0.48–1.93; P =0.92; $I^2 = 29\%$) (14, 27), respectively, corresponding to the above mentioned symptoms.

Other AEs were also collected such as hyocalcemia, anemia, arthralgia, pain in extremity, headache, edema peripheral, cough, neutropenia, musculoskeletal pain, dyspnoea, grade 3-4 AEs, dizziness and back pain. Based on the extracted data, we conducted meta-analyses with corresponding summary RR results of 0.91 (95% CI 0.80–1.03; P = 0.14; $I^2 = 0\%$) (23, 27), 0.98 (95% CI 0.68–1.42; P = 0.91; $I^2 = 38\%$) (22, 25, 27), 0.82 $(95\% \text{ CI } 0.62-1.08; P = 0.15; I^2 = 0\%) (14, 25, 27), 0.88 (95\% \text{ CI})$ $0.64-1.21; P = 0.44; I^2 = 0\%$ (14, 25, 27), 1.00 (95% CI 0.69-1.43; $P = 0.98; I^2 = 0\%$ (14, 25, 27), 0.93 (95% CI 0.62–1.39; P = 0.73; $I^2 = 0\%$ (14, 25, 27), 1.23 (95% CI 0.80–1.87; P = 0.34; $I^2 = 0\%$) $(14, 25, 27), 0.55 (95\% \text{ CI } 0.28-1.10; P = 0.09; I^2 = 1\%) (14, 27),$ 0.95 (95% CI 0.81–1.10; P = 0.50; $I^2 = 0\%$), 0.80 (95% CI 0.54– 1.20; P = 0.29; $I^2 = 0\%$) (14, 25) and 0.89 (95% CI 0.57–1.40; P =0.64; $I^2 = 0\%$) (14, 27). For dizziness and back pain, de-escalated therapy could induce significant decreases (RR 0.52 95% CI 0.32-0.86; P = 0.01; $I^2 = 70\%$ and RR 0.69 95% CI 0.49-0.96; P = 0.03; $I^2 = 0\%$) (25, 27).

In addition, the study by Himelstein et al. in which all patients received daily supplementation with calcium and vitamin D, demonstrated that the rate of hypocalcemia driven by zoledronate was 32.7% for the de-escalated group and 35.3% for the standard group with no statistical difference (P = 0.25), respectively (23).

Biomarkers of Bone Turnover

The change in biomarker levels on study was assessed, given that they have been used in clinical trials as a surrogate marker of bone metastases control and hence BPs efficacy (28-30). Four studies reported biomarkers of bone turnover in the form of CTx and urinary N-terminal telopeptide (NTx). Addison et al. reported that, compared to the standard group, patients in the de-escalated group had statistically significantly greater increases in CTx levels (median of 131 vs. 17, P = 0.034) and no statistically significant change in NTx (median of 45 vs. 43, P = 0.54) (21). The ZOOM study assessed median percentage change in NTx concentration from baseline and indicated it was significantly higher in the de-escalated group vs. the standard group at 6 months (12.2 vs. -2.3%; P = 0.011), 9 months (10.6 vs. -2.2%; P = 0.047), and 12 months (12.2 vs. 0.0%; P = 0.047) (14). The REFORM study showed a consistent and maintained increase in absolute CTx levels in women receiving de-escalated therapy,





De-escalated Standard Mean Difference Mean Difference Study or Subgroup Mean SD Total Mean SD Total Weight IV, Fixed, 95% CI IV, Fixed, 95% CI Amadori 2013 0.26 0.81 216 0.22 0.6 209 3.7% 0.04 [-0.10, 0.18] Himelstein 2017 0.4 0.29 911 0.4 0.29 0.00 [-0.03, 0.03] 911 95.3% Hortobagyi 2017 203 0.46 1.06 200 1.1% 0.04 [-0.21, 0.29] 0.5 1.5 Total (95% CI) 1320 100.0% 0.00 [-0.02, 0.03] 1330 Heterogeneity: Chi² = 0.41, df = 2 (P = 0.81); l² = 0% -0.1 0.1 0.2 -0.2 Ó Test for overall effect: Z = 0.14 (P = 0.89) Favours [De-escalated] Favours [Standard] FIGURE 5 | Meta-analysis result of SMR.



which met the prespecified cutoff for clinical significance (P = 0.096) (22). Himelstein et al. made a descriptive evaluation of CTx measurement results (23). The observed CTx levels were higher at each time point among patients receiving the deescalated zoledronate but involved no statistical test. Given the different scales in the reporting of outcomes, a meta-analysis was not carried out.

DISCUSSION

The rational dosing interval of BPs has gradually attracted attention especially to oncologists and patients with bone metastases, due to the increased toxicity caused by a long-term medication (31, 32). Along with the advancing clinical studies being designed to explore its optimal dosing frequency, it is of the utmost importance to gain a better understanding of the

fact that whether a less intensive dosing schedule of BPs therapy might provide the comparable efficacy as the standard one while reducing side effects (33).

First of all, as the most commonly used efficacy evaluated outcome, SRE-related data was available in a total of six studies, in terms of SRE rate, SMR and time-to-first SRE. The metaanalysis result of the number of patients having an on-study SRE showed evidence that a longer-interval BPs regimen (Q12w) was similar to a standard schedule (Q3-4w) toward patients with bone metastases. Meanwhile, there was no evidence of a difference in SMR between the de-escalated and standard groups, with an inconclusive CI and summary estimate near the null (RR 0.00, 95% CI -0.02 - 0.03; P = 0.89; $I^2 = 0\%$). Two articles reported time-to-first SRE and showed no significant statistical difference between the two treatments (14, 25). Unfortunately, one of the articles only stated descriptive results. There was no sufficient, consistent data for a meta-analysis to be carried out. Furthermore, individual types of SREs were analyzed, including radiation to bone, clinical fractures, spinal cord compression and surgery involving bone. It is noteworthy that the de-escalated regimen led to a significantly increased incidence of surgery involving bone compared with the standard care, despite the comparable result of the general SRE rate. Consequently, we consider this to be a point of concern in evaluating the efficacy of both treatments.

Secondly, following a thorough review of toxicity, there were new safety findings according to the meta-analysis results. The proportion of patients who experienced at least one AE revealed a slightly decreased tendency in the longer-interval dosing regimen, although this was not statistically significant. Nevertheless, de-escalated regimen did provide an overall benefit by reducing the risk of gastrointestinal disorders, back pain and dizziness by 27, 31, and 48%, respectively, comparing to the standard therapy (**Figure 6B**, **Figures S3**, **S4**). Besides, the de-escalated dosing of BPs slightly depressed the risk of acute-phase reactions (such as fatigue and nausea), although there was no statistical significance (**Figure S1**).

For on-study renal dysfunction and ONJ, different dosing frequencies with BPs did not significantly affect the occurrence rate, although data from individual studies would suggest that standard dosing was associated with an increased rate (23, 25). However, it needs to be pointed out that as the most serious type of AEs, ONJ occurs after cumulative dosing of BPs, and it is therefore likely that some included studies with limited follow-up period would not allow for the detection of such event (34–36). Longer-term studies might be necessary to show any benefit from de-escalated dosing on the occurrence of ONJ.

Furthermore, circulating telopeptide levels have been used as endpoints to determine drug efficacy and an indicator for the relationship with SRE risk as well (37, 38). The measurement index for turnover biomarkers varied among studies, by using different indicators such as NTx and CTx, hence data extracted was not consistent enough for pooled analysis. In general, the observed levels of bone turnover biomarkers were higher at each time point among patients receiving BPs in de-escalated groups. In the ZOOM study and the Addison study, there is even a significant difference in median percentage change from baseline in NTx and CTx concentration in the 4-week group vs. the 12-week group (14, 21). The recovery of bone turnover is probably a result of the rapid removal of stored BPs through increased osteoclast activity at sites of active bone lesions (16, 39). Despite this, given that an increasing incidence of surgery involving bone was directly related to the de-escalated regimen, the significance of certain biomarkers should be highly stressed in the future work. These findings raise questions around the longterm efficacy of de-escalation of bone-targeted therapy as well. On the other hand, the available data of recent work highlighted the need for completion and analysis of larger definitive trials to assess the roles of biomarkers as predictors of SRE risk.

This review has a few limitations that need to be addressed. First, there was clinical and methodological heterogeneity among

studies in terms of duration of studies and different schedules of prior BPs usage. Secondly, one study terminated early due to challenges with enrollment, achieving only 289 of the targeted 1,400 (24); as such, it was unclear whether study findings might have changed with additional recruitment. Thirdly, only trials among MBC, MPC, or MM patients were available to be analyzed. Therefore, further studies are still warranted to make a final conclusion as to whether the results can be generalized to other types of cancers. Finally, the study did not address any survival data, risks or benefits of administering BPs for more than 2 years. As tumor bone metastases are equivalent to chronic diseases that require long-term treatment of BPs, corresponding follow-up data of more than 2 years should be necessary. Despite these limitations, it is of interest to note that there were consistent patterns across all trials, all of which were multi-center studies and most of which involved multi-ethnic patients. These findings confirmed that a longer-interval of BPs was similar to the standard therapy in reducing the risk of developing SREs. What's more, the deescalated regimen was proven for the first time to provide an overall safety benefit by reducing the risk of gastrointestinal disorders and significantly relieving the phenomenon of back pain.

CONCLUSIONS

For malignant tumor patients with bone metastases, a deescalated BPs strategy is proven to have a better safety profile compared to standard dosing. Although the efficacy is generally comparable on SRE and SMR between the two dosing regimens, trials with long durations and large sample sizes are still warranted to make a solid judgment.

DATA AVAILABILITY

All datasets generated for this study are included in the manuscript and/or the **Supplementary Files**.

AUTHOR CONTRIBUTIONS

QL and PM conceived the study and collected data. ZL analyzed the data. SZ and MJ verified the data. QL, PM, and ZL prepared and edited the manuscript. All authors contributed to the paper and perspectives, and agreed with its final content.

FUNDING

This work was financially supported by the National Nature Science Foundation of China (No. 81803442, No. 81872939).

SUPPLEMENTARY MATERIAL

The Supplementary Material for this article can be found online at: https://www.frontiersin.org/articles/10.3389/fonc. 2019.00774/full#supplementary-material

REFERENCES

- 1. Coleman RE. Adjuvant bone-targeted therapy to prevent metastasis: lessons from the AZURE study. *Curr Opin Support Palliat Care.* (2012) 6:322–9. doi: 10.1097/SPC.0b013e32835689cd
- Holen I, Coleman RE. Bisphosphonates as treatment of bone metastases. Curr Pharm Des. (2010) 16:1262–71. doi: 10.2174/138161210791034003
- Coleman RE. Metastatic bone disease: clinical features, pathophysiology and treatment strategies. *Cancer Treat Rev.* (2001) 27:165–76. doi: 10.1053/ctrv.2000.0210
- Mundy GR. Is prolonged stimulation of bone growth a therapeutic possibility? Mol Cell Endocrinol. (1991) 75:C19–25. doi: 10.1016/0303-7207(91)90222-E
- Rogers MJ, Watts DJ, Russell RG. Overview of bisphosphonates. Cancer. (1997) 80(8 Suppl):1652–60.
- 6. Rosen LS, Gordon D, Kaminski M, Howell A, Belch A, Mackey J, et al. Zoledronic acid versus pamidronate in the treatment of skeletal metastases in patients with breast cancer or osteolytic lesions of multiple myeloma: a phase III, double-blind, comparative trial. *Cancer J.* (2001) 7:377–87.
- Clemons M, Gelmon KA, Pritchard KI, Paterson AH. Bone-targeted agents and skeletal-related events in breast cancer patients with bone metastases: the state of the art. *Curr Oncol.* (2012) 19:259–68. doi: 10.3747/co.19.1011
- Kuchuk I, Hutton B, Moretto P, Ng T, Addison CL, Clemons M. Incidence, consequences and treatment of bone metastases in breast cancer patients experience from a single cancer centre. *J Bone Oncol.* (2013) 2:137–44. doi: 10.1016/j.jbo.2013.09.001
- O'Carrigan B, Wong MH, Willson ML, Stockler MR, Pavlakis N, Goodwin A. Bisphosphonates and other bone agents for breast cancer. *Cochrane Database Syst Rev.* (2017) 10:CD003474. doi: 10.1002/14651858.CD003474.pub4
- Body JJ, Bartl R, Burckhardt P, Delmas PD, Diel IJ, Fleisch H, et al. Current use of bisphosphonates in oncology. International bone and cancer study group. J Clin Oncol. (1998) 16:3890–9. doi: 10.1200/JCO.1998.16.12.3890
- National Comprehensive Cancer Network: NCCN Clinical Practice Guidelines in Oncology: Multiple Myeloma. V2 (2016). Available from: http://www.nccn. org/professionals/physician_gls/pdf/myeloma.pdf
- Kyle RA, Yee GC, Somerfield MR, Flynn PJ, Halabi S, Jagannath S, et al. American Society of Clinical Oncology 2007 clinical practice guidelineupdate on the role of bisphosphonates in multiple myeloma. *J Clin Oncol.* (2007) 25:2464–72. doi: 10.1200/JCO.2007.12.1269
- Durie BG. Use of bisphosphonates in multiple myeloma: IMWG response to Mayo Clinic consensus statement. *Mayo Clin Proc.* (2007) 82:516–17; author reply 517–18. doi: 10.4065/82.4.516
- Amadori D, Aglietta M, Alessi B, Gianni L, Ibrahim T, Farina G, et al. Efficacy and safety of 12-weekly versus 4-weekly zoledronic acid for prolonged treatment of patients with bone metastases from breast cancer (ZOOM): a phase 3, open-label, randomised, non-inferiority trial. *Lancet Oncol.* (2013) 14:663–70. doi: 10.1016/S1470-2045(13)70174-8
- Kimmel DB. Mechanism of action, pharmacokinetic and pharmacodynamic profile, and clinical applications of nitrogen-containing bisphosphonates. J Dent Res. (2007) 86:1022–33. doi: 10.1177/154405910708601102
- Addison CL, Bouganim N, Hilton J, Vandermeer L, Dent S, Amir E, et al. A phase II, multicentre trial evaluating the efficacy of de-escalated bisphosphonate therapy in metastatic breast cancer patients at low-risk of skeletal-related events. *Breast Cancer Res Treat.* (2014) 144:615–24. doi: 10.1007/s10549-014-2906-x
- Ibrahim MF, Mazzarello S, Shorr R, Vandermeer L, Jacobs C, Hilton J, et al. Should de-escalation of bone-targeting agents be standard of care for patients with bone metastases from breast cancer? A systematic review and meta-analysis. *Ann Oncol.* (2015) 26:2205–13. doi: 10.1093/annonc/mdv284
- Jacobs C, Ng T, Ong M, Clemons M. Long-term benefits versus sideeffects from bone-targeted therapies for cancer patients: minimizing risk while maximizing benefits. *Curr Opin Support Palliat Care.* (2014) 8:420–8. doi: 10.1097/SPC.00000000000084
- Green SB, Pappas AL. Effects of maternal bisphosphonate use on fetal and neonatal outcomes. Am J Health Syst Pharm. (2014) 71:2029–36. doi: 10.2146/ajhp140041
- 20. Liberati A, Altman DG, Tetzlaff J, Mulrow C, Gotzsche PC, Ioannidis JP, et al. The PRISMA statement for reporting systematic reviews and meta-analyses of studies that evaluate health care interventions:

explanation and elaboration. Ann Intern Med. (2009)151. W65-94. doi: 10.7326/0003-4819-151-4-200908180-00136

- Addison CL, Pond GR, Zhao H, Mazzarello S, Vandermeer L, Goldstein R, et al. Effects of de-escalated bisphophonate therapy on bone turnover biomarkers in breast cancer patients with bone metastases. *Springerplus*. (2014) 3:577. doi: 10.1186/2193-1801-3-577
- 22. Amir E, Freedman O, Carlsson L, Dranitsaris G, Tomlinson G, Laupacis A, et al. Randomized feasibility study of de-escalated (every 12 wk) veresus standard (every 3 to 4 wk) intravenous pamidronate in women with low-risk bone metastases from breast cancer. *J Clin Oncol.* (2013) 36:436–42. doi: 10.1097/COC.0b013e3182568f7a
- Himelstein AL, Foster JC, Khatcheressian JL, Roberts JD, Seisler DK, Novotny PJ, et al. Effect of longer-interval vs standard dosing of zoledronic acid on skeletal events in patients with bone metastases: a randomized clinical trial. *JAMA*. (2017) 317:48–58. doi: 10.1001/jama.20 16.19425
- 24. Coleman RE, Wright J, Houston S, Agrawal R, Purohit PK. Randomized trial of marker-directed versus standard schedule zoledronic acid for bone metastases from breast cancer. *J Clin Oncol.* (2012) 30:511.
- 25. Hortobagyi GN, Van Poznak C, Harker WG, Gradishar WJ, Chew H, Dakhil SR, et al. Continued treatment effect of zoledronic acid dosing every 12 vs 4 weeks in women with breast cancer metastatic to bone: the OPTIMIZE-2 randomized clinical trial. *JAMA Oncol.* (2017) 3:906–12. doi: 10.1001/jamaoncol.2016.6316
- Kuchuk I, Beaumont J, Clemons M, Amir E4, Addison CL, Cella D. Effects of de-escalated bisphosphonate therapy on the functional assessment of cancer therapy-bone pain, brief pain inventory and bone biomarkers. *J Bone Oncol.* (2013) 24:154–7. doi: 10.1016/j.jbo.2013.07.004
- ClinicalTrials.gov Identifier: NCT00424983. (2010). Available online at: https://clinicaltrials.gov/ct2/show/NCT00424983; https://www.novctrd.com/ CtrdWeb/searchbystudyid.nov?studyId=CZOL446E2105
- Clamp A, Danson S, Nguyen H, Cole D, Clemons M. Assessment of therapeutic response in patients with metastatic bone disease. *Lancet Oncol.* (2004) 5:607–16. doi: 10.1016/S1470-2045(04)01596-7
- 29. Clemons MJ, Cochrane B, Pond GR, Califaretti N, Chia SK, Dent RA, et al. Randomised, phase II, placebocontrolled, trial of fulvestrant plus vandetanib in postmenopausal women with bone only or bone predominant, hormone-receptor-positive metastatic breast cancer (MBC): the OCOG ZAMBONEY study. *Breast Cancer Res Treat.* (2014) 146:153–62. doi: 10.1007/s10549-014-3015-6
- Brown JE, Thomson CS, Ellis SP, Gutcher SA, Purohit OP, Coleman RE. Bone resorption predicts for skeletal complications in metastatic bone disease. Br J Cancer. (2003) 89:2031–7. doi: 10.1038/sj.bjc. 6601437
- Hutton B, Addison C, Mazzarello S, Joy AA, Bouganim N, Fergusson D, et al. De-escalated administration of bonetargeted agents in patients with breast and prostate cancer-a survey of Canadian oncologists. *J Bone Oncol.* (2013) 2:77–83. doi: 10.1016/j.jbo.2013.03.001
- Hutton B, Morretto P, Emmenegger U, Mazzarello S, Kuchuk I, Addison CL, et al. Bone-targeted agent use for bone metastases from breast cancer and prostate cancer: a patient survey. J Bone Oncol. (2013) 2:105–9. doi: 10.1016/j.jbo.2013.05.002
- Kuchuk I, Clemons M, Addison C. Time to put an end to the 'one size fits all' approach to bisphosphonate use in patients with metastatic breast cancer? *Curr Oncol.* (2012) 19: e303–4. doi: 10.3747/co.19.1009
- Fung PL, Nicoletti P, Shen Y, Porter S, Fedele S. Pharmacogenetics of Bisphosphonate-associated Osteonecrosis of the Jaw. Oral Maxillofac Surg Clin North Am. (2015) 27:537–46. doi: 10.1016/j.coms.2015.06.005
- Tsao C, Darby I, Ebeling PR, Walsh K, O'Brien-Simpson N, Reynolds E, et al. Oral health risk factors for bisphosphonate-associated jaw osteonecrosis. J Oral Maxillofac Surg. (2013) 71:1360–6. doi: 10.1016/j.joms.2013.02.016
- Khosla S, Bilezikian JP, Dempster DW, Lewiecki EM, Miller PD, Neer RM, et al. Benefits and risks of bisphosphonate therapy for osteoporosis. J Clin Endocrinol Metab. (2012) 97:2272–82. doi: 10.1210/jc. 2012-1027
- 37. Raje N, Vescio R, Montgomery CW, Badros A, Munshi N, Orlowski R, et al. Bone marker-directed dosing of zoledronic acid for the prevention of skeletal complications in patients with multiple myeloma: results of the Z-MARK

study. Clin Cancer Res. (2016) 22:1378-84. doi: 10.1158/1078-0432.CCR-15-1864

- Lipton A, Fizazi K, Stopeck AT, Henry DH, Smith MR, Shore N, et al. Effect of denosumab versus zoledronic acid in preventing skeletal-related events in patients with bone metastases by baseline characteristics. *Eur J Cancer*. (2016) 53:75–83. doi: 10.1016/j.ejca.2015.09.011
- Mulcahy LE, Curtin CM, McCoy RJ, O'Brien FJ, Taylor D, Lee TC, et al. The effect of bisphosphonate treatment on the biochemical and cellular events during bone remodelling in response to microinjury stimulation. *Eur Cell Mater.* (2015) 30:271–81. doi: 10.22203/eCM. v030a19

Conflict of Interest Statement: The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Copyright © 2019 Luo, Men, Liu, Zhai and Jiang. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.